Author’s response to reviews

Title: From prospective clinical trial to reducing social inequalities in health: concept and design of a multidisciplinary study in precarious patients with breast cancer. The DESSEIN trial.

Authors:

Charlotte Ngo (charlotte.ngo@aphp.fr)
Aurélia Dinut (aurelia.dinut@aphp.fr)
Audrey Bochaton (bochaton@gmuil.com)
Hélène Charreire (helene.charreire@u-pec.fr)
Caroline Desprès (caroline.despres@bbox.fr)
Sandrine Baffert (sandrine.baffert@cemka.fr)
Fabrice Lécuru (fabrice.lecuru@aphp.fr)
Gilles Chatellier (gilles.chatellier@aphp.fr)

Version: 2 Date: 06 Sep 2019

Author’s response to reviews:

Dear Editor, thank you for reviewing our manuscript.

We added in the Funding statement of the declarations that the funding body played no role in the design, nor in the data collection, analysis and interpretation, and no role in writing the manuscript.

Best regards

Charlotte Ngo